Author’s response to reviews

Title: Bacteriologically confirmed extra pulmonary tuberculosis and treatment outcome of patients consulted and treated under program conditions in the littoral region of Cameroon

Authors:

Pride Mbuh Teyim (teyimpride@yahoo.com)
Irene Ane-Anyangwe (ianyangwe@yahoo.com)
Wandji Adeline (adywandji@gmail.com)
Benjamin D. Thumamo Pokam (thumamo@yahoo.fr)
Henry Dilonga Meriki (merilonga@yahoo.com)
Wilfred Fon Mbacham (wfmbacham@yahoo.com)

Version: 1 Date: 03 Sep 2018

Author’s response to reviews:

figure titles and captions from the figure files have been removed,

The data sets used and/or analyzed during the current study are available from the corresponding author on reasonable request.